facsimile to (202) 395–7285. Comments may also be submitted to the Department of Justice (DOJ), Justice Management Division, Information Management and Security Staff, Attention: Department Clearance Officer, Suite 850, 1001 G Street, N.W., Washington, D.C. 20530. Additionally, comments may be submitted to DOJ via facsimile to (202) 514–1534.

The proposed collected is listed below.

- (1) Type of information collection. Reinstatement, with no change, of a previously approved collection for which approval has expired.
- (2) The title of the form/collection. Victims of Crime Act, Victim Assistance Grant Program, Performance Report.
- (3) The agency form number if any, and the applicable component of the Department sponsoring the collection. Form: OJP Admin. Form 7330 (Rev. 11/95)

Office for Victims of Crime, Office of Justice Programs, U.S. Department of Justice

- (4) Affected public who will be asked or required to respond, as well as a brief abstract. Primary: State government Other: None
- (5) An estimated total of the number of respondents and the amount of time estimated for an average respondent to respond. The information to compile these reports will be drawn from victim assistance program data to the 57 respondents (grantees). The number of victim assistance vary widely from state to state. A state could be responsible for compiling subgrant data for as many as 186 programs (Texas) to as few as four programs (Guam). Therefore, the estimated clerical hours can range from 1 to 70 hours.
- (6) An estimate of the total burden (in hours) associated with the collection: The current estimated burden is 1,197 (20 hours per respondent (estimated median) + 1 hour per respondent for recordkeeping x 57 respondents = 1,197 hours). There is no increase in the annual recordkeeping and reporting burden.

If additional information is required contact: Mr. Robert B. Briggs, Clearance Officer, United States Department of Justice, Information Management and Security Staff, Justice Management Division, Suite 850, Washington, D.C. 20530.

Robert B. Briggs,

Department Clearance Officer, United States Department of Justice.

[FR Doc. 98–33132 Filed 12–14–98; 8:45 am] BILLING CODE 4410–18–M

DEPARTMENT OF LABOR

Employment and Training Administration

Notice of a Change in Status of an Extended Benefit (EB) Period for Puerto Rico

This notice announces a change in benefit period eligibility under the EB Program for Puerto Rico.

Summary

The following change has occurred since the publication of the last notice regarding the State's EB status:

• November 1, 1998: Puerto Rico triggered "on" EB. Puerto Rico's 13-week insured unemployment rate rose above the 6.0 percent threshold necessary to be triggered "on" to EB for the week ending October 17, 1998.

Information for Claimants

The duration of benefits payable in the EB Program, and the terms and conditions on which they are payable, are governed by the Federal-State Extended Unemployment Compensation Act of 1970, as amended, and the operating instructions issued to the States by the U.S. Department of Labor. In the case of a State beginning an EB period, the State employment security agency will furnish a written notice of potential entitlement to each individual who has exhausted all rights to regular benefits and is potentially eligible for EB (20 CFR 615.13(c)(1)).

Persons who believe they may be entitled to EB, or who wish to inquire about their rights under the program, should contact the nearest State employment service office or unemployment compensation claims office in their locality.

Signed at Washington, DC, on December 7, 1998.

Raymond Bramucci,

Assistant Secretary of Labor for Employment and Training.

[FR Doc. 98–33188 Filed 12–14–98; 8:45 am]

DEPARTMENT OF LABOR

Employment and Training Administration

Job Training Partnership Act, Title III, Demonstration Program: Dislocated Worker Manufacturing Technology Demonstration Program

AGENCY: Employment and Training Administration, Labor.

ACTION: Notice of Availability of Funds and Solicitation for Grant Applications (SGA).

SUMMARY: All information required to submit a grant application is contained in this announcement. The U.S. Department of Labor (DOL), **Employment and Training** Administration (ETA), announces a demonstration program to test the ability of the workforce development system to partner with employers, training providers and others to train dislocated workers in the skills necessary to obtain work requiring technology skills in occupations in manufacturing industry settings with long-term growth potential. The program will be funded with Secretary's National Reserve funds appropriated for Title III of the Job Training Partnership Act (JTPA) and administered in accordance with 29 CFR part 95 and 97 as applicable.

This notice provides information on the process that eligible entities must use to apply for these demonstration funds and how grantees will be selected. It is anticipated that up to \$10 million will be available for funding demonstration projects covered by this solicitation, with no award being more than \$1 million.

DATES: The closing date for receipt of proposals is February 16, 1999 at 4 p.m. (Eastern Time).

ADDRESSES: Applications shall be mailed to: U.S. Department of Labor; Employment and Training Administration; Division of Acquisition and Assistance; Attention: Yvonne Harrell, Reference: SGA/DAA 99–001; 200 Constitution Avenue, NW., Room S–4203; Washington, DC 20210.

FOR FURTHER INFORMATION CONTACT: Fax questions to Yvonne Harrell, Division of Acquisition and Assistance at (202) 219–8739 (this is not a toll-free number).

SUPPLEMENTARY INFORMATION: This announcement consists of five parts. Part I describes the authorities and purpose of the demonstration program and identifies demonstration policy. Part II describes the application process and provides guidelines for use in applying for demonstration grants. Part III includes the statement of work for the demonstration projects. Part IV describes the selection process, including the criteria that will be used in reviewing and evaluating applications. Part V discusses the demonstration program's monitoring, reporting and evaluation.

Part I. Background

A. Authorities

Section 323(a)(6) of JTPA (29 U.S.C. 1662b) authorizes the use for demonstration programs of funds reserved under section 302 of JTPA (29 U.S.C. 1652) and provided by the Secretary for that purpose under section 322 of JTPA (29 U.S.C. 1662a). Demonstration program grantees must comply with all applicable federal and state laws and regulations in setting up and carrying out their programs.

B. Purpose

It is now well understood that the economy has transitioned from the industrial age to the information age. In this age, the most valued commodities are the skills and knowledge possessed by the individual. Global competition has reached an unprecedented level. Technology plays an increasingly important role in this global scenario as nations strive to build things or provide services that are faster, better or cheaper than their competitors.

In this era of global competition and rapid technological advances, technology is the most critical driver of economic growth. The U.S. Department of Commerce, Office of Technology Policy, has reported advances in technology to be the single most important determining factor in sustaining economic growth, estimated to account for as much as half of the Nation's long-term economic growth over the past 50 years. Technology as a method for achieving a practical result encompasses the techniques, machines and equipment, controls, processes, and organization of work, as well as the ideas, skills, and knowledge underlying the work process. Traditional work environments have altered, as have the requisite skills needed by workers to succeed in today's workplace. Technology provides the tools for creating a wide array of new and improved products and new services that reach well beyond the narrow confines of traditional labor markets. A product can now be provided from almost any community, each with the potential to reach global markets. The ability of a company to innovate, incorporate technology, improve products or services, increase market share and thus expand capacity and employment is the engine of economic growth.

Information technologies affect almost every sector and every industry in the United States, in terms of digitally based products, services, and production and work processes. The very nature of advanced technology lies in the ability

of a business or industry to identify, assess, adopt and incorporate information based technologies into everyday business and production processes. The information/knowledgebased workplace of today's leading companies requires workers to possess conceptual, analytical, communication, interpersonal, and self-management skills beyond the basic academic and technical skills of the traditional workplace. There is often a skills deficit experienced by employers who continuously push the envelope to innovate, and adopt new technology in order to stay ahead of competitors, both domestic and international.

With accelerated changes in technology, America's workers often discover their skill base has become out of date. New approaches are needed to help American workers stay competitive. Workers need to know and understand what skill standards employers are looking for, and they need to have the means to raise their skills to match that demand.

Our Nation's workforce development system is working to meet this need, but skill shortages in technology are currently very high in some industry sectors and geographic areas. Severe shortages of workers who can apply and use advanced technologies could undermine U.S. innovation, productivity, and competitiveness in world markets. A steady supply of skill workers will help our Nation's industries remain competitive. More importantly, these workers need to possess the appropriate skills demanded in the workplace. Ideally, a system of "just in time" training would be able to supply skilled workers that meet industry driven standards and certifications.

The purpose of this demonstration is to test the ability of the Nation's workforce development system to partner with employers, training providers and others to train dislocated workers in the skills necessary to obtain work requiring technology skills in manufacturing occupations and industries experiencing shortages of such workers.

Industries such as aerospace, computers and electronics manufacturing, machinery and motor vehicles, chemicals and petroleum, and specialized instruments and devices as well as bio-technical/biomedical could be among the manufacturing industries experiencing technology skill shortages among those workers they seek to employ.

Manufacturing technologies have gone through several metamorphoses. The impact of these technological trends

is often felt as a loss in the number of unskilled jobs with an increase in more technology-savvy jobs required to control automated, computer-operated machinery. As the tools and equipment become more automated, the skills needed for entry level technicians and operators multiply. Increasing use of robotics requires employees who once performed manual labor to become technicians who control automatic processes remotely by computer. Assemblers frequently are now required to possess computer skills for controlling automatic processes remotely. The shift towards automating the production line has resulted in a need for workers who are able to work with computers, robotics, and Computer Numerically Controlled (CNC) machines. Instead of interacting with the products, line workers may now manage machines which perform the processes.

A large number of the layoffs that occur within a manufacturing company are associated with elimination or reduction of a product and changes in technology. They generally have the largest impact on those with lower or outdated skills. Amid massive and continual restructuring within manufacturing, it is imperative for manufacturing employees to commit to lifelong learning. The technology and the push to produce a particular product that created their present position is often soon to be replaced by another generation of product and production process.

As a part of the Nation's Workforce Development System, programs presently funded under Title III of the Job Training Partnership Act, and soon to be funded under the Workforce Investment Act of 1998, annually provide adjustment and training assistance to over 500,000 individuals who have lost their jobs through no fault of their own. These employment and training funds targeted to serve dislocated workers are managed through State and local workforce development organizations who design and operate a national system for training and reemployment programs based on: (1) The needs and characteristics of the local dislocated worker population; (2) the needs of local employers for skilled workers; and (3) the capabilities and capacities of training institutions and other local service providers. The emerging infrastructure of One-Stop/ Career Center systems provides comprehensive and integrated workforce development services to both participants and employers.

Under this demonstration, the Department will fund projects that

specifically document the existence of and respond to the reported shortages in their geographical area of workers in manufacturing jobs requiring technology skills. For purposes of this solicitation, the term "technology" may be viewed broadly as the link between people and technology in the workplace.

Successful applications may be based on the use of new or innovative service strategies such as the involvement of under represented groups of dislocated workers for existing training programs; the development and use of curricula geared specifically to eligible groups of dislocated workers and the needs of employers with openings in technologyrelated jobs; or the development of concentrated training models for workers with a residue of skill knowledge from previous related employment, or use of curriculum and skills training interventions designed to impart knowledge, skills and abilities of industry skill standards (where available or under development).

Each successful application will document where there are strong linkages with specific employers' demand for workers with technology-related skills. The demonstration program goals of placement of the project participants in jobs using technology in manufacturing industries which are targeted in the proposal must be clearly addressed and sufficient assurance must be demonstrated that this goal can be accomplished.

Participant satisfaction with project services and with their jobs, as well as their employer's satisfaction with project services and with the participants' skill level and work, should be measured not only at the end of the project but also at critical points identified by the applicant during the progress of the demonstration's implementation.

C. Demonstration Policy

1. Grant Awards

DOL anticipates awarding ten (10) to fifteen (15) grants, not to exceed \$1 million per grant. It is anticipated that awards will be made by April 30, 1999. Award decisions will be published on the Internet at ETA's Home Page at http://www.doleta.gov.

2. Eligible Applicants

Any organization capable of fulfilling the terms and conditions of this solicitation may apply. Applicants who are not Substate grantees under JTPA Title III, or One-Stop Career Center Operators or Workforce Investment Boards under WIA must submit a letter from the authorized signatory of one or

more of such organizations continuing comments on the applicant's proposal. Under Lobbying Disclosure Act of 1995, section 18, an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 which engages in lobbying activities shall not be eligible for the receipt of Federal funds constituting an award grant or loan. This is a risk free Federal program: Therefore, all for profit organizations that apply will not be able to receive a fee if awarded a grant.

3. Eligible Participants

All participants in projects funded under this demonstration program must be either:

- (a) Eligible dislocated workers as defined at JTPA section 301(a)(1), and 314(h)(1) of the Job Training Partnership Act. These sections of the law may be viewed at http://doleta.gov/regs/ statutes/jtpalaw.htm. Proposed projects may target subgroups of the eligible population based on factors such as (but not limited to) occupation, industry, nature of dislocation, and reason for unemployment. Note: Individuals whose eligibility is based upon their status as long-term unemployed (section 301(a)(1)(C) JTPA) must have a demonstrated attachment to the labor force.
- (b) Incumbent workers. These are currently-employed workers whose employers have determined that the workers require training in order to help keep their firms competitive and the subject workers employed, avert layoffs, upgrade workers' skills, increase wages earned by employees and/or keep workers skills competitive. Such technology training would support further job retention and career development for improved economic self-sufficiency for employed workers, especially those most vulnerable to job loss, and increase the capability of the employing firm(s) to access and retain skilled workers.

4. Applicable Technology

Applicant's proposal must describe the technology skills to be demonstrated in the grant in the context of the skills presently in use in the industry or plants—e.g., how this demonstration is related to the introduction of new equipment, upgrading incumbent workers, development of a new product. If this technology application will enable improvements in the manufacturing process, a description of such benefits should be provided. If this technology is linked to a specific employer or group of employers, discuss the impact on present skill levels caused by the demonstration activities including changes caused by

equipment, materials or work organization. Where applicable, relate changes to factors affecting workers such as increased or decreased decision-making responsibility, changes in advancement opportunities or transferability of new skills, changes in the pace of work, and wage increases related to increased skill attainment. Indicate whether this technology could be considered "leading edge" by the industry.

5. Allowable Activities

Funds provided through this demonstration may be used only to provide services of the type described at section 314(c) and (d) of JTPA. Supportive services may be provided when they are necessary to enable an individual who is eligible for training but cannot afford to pay for such supportive services, to participate in the training program. These services are defined in section 4(24) of JTPA. (Use ETA's web site reference above to view.)

Grant funds may be used to reimburse employers for extraordinary costs associated with on-the-job training of program participants, in accordance with the provisions of 20 CFR 627.240. In addition to the limitations and requirements provided in JTPA, particularly at Part C of Title I, prospective applicants should be aware that grant funds may not be used for the following purposes: (a) For training that an employer is in a position to provide and would have provided in the absence of the requested grant; (b) to pay salaries for program participants; and (c) for acquisition of production equipment. Applicants may budget limited amounts of grant funds to work with technical experts or consultants to provide advice and develop more complete project plans after a grant award, however, the level of detail in the project plan may affect the amount of funding provided.

Grant activities may include: (a) Development, testing and initial application of curricula focused on intensive, short-term training to get participants into productive, high demand information or advanced technology employment as quickly as possible;

(b) Working with employers in develop and apply worksite-based learning strategies that utilize cuttingedge technology and equipment;

(c) Development of employer-based training programs that will take advantage of opportunities created by employers' needs for workers with new technology skills;

(d) Development and initial application of contextual learning opportunities for participants to learn

technology theory in a classroom setting while applying that learning in an on-

the-job setting;

(e) Use of curriculum and skills training programs that are designed to impart learning to meet employerspecified or industry specific skill standards or certification requirements;

(f) Convening of an Employer Advisory Board to identify skills gaps of job applicants and present workers affecting the ability of the employer to offer a competitive product and develop a strategy for retraining.

a strategy for retraining;

(g) Innovative linkage and collaboration between employers and the local Substate Grantee and/or One-Stop/Career Center system to ensure a steady supply of high demand, high skill information or advanced technology workers.

The above are illustrative examples and are not intended to be an exhaustive listing of possible demonstration project designs or approaches which may achieve the purpose of this solicitation. However, successful applicants must demonstrate the direct involvement by employers experiencing skill shortages in the design and operation of the project as well as provide substantive documentation about the existence of skill shortages for the industry or occupations to be targeted by the proposed project. Documentation should include a description of the employer involvement anticipated in the project. An employer advisory committee may be one means of accomplishing employer involvement.

6. Coordination

In order to maximize the use of public resources and avoid duplication of effort, applicants will coordinate the delivery of services under this demonstration with the delivery of services under other programs (public or private), available to all or part of the target group. Projects linking or collaborating with an existing USDOL funded One-Stop/Career Center initiative and/or local JTPA Substate Grantee located within a project area fulfill this requirement. The use of Pell Grants for eligible workers or the use of State training or education funds provided for dislocated workers or certain types of employers should also be addressed in the application.

7. Period of Performance

The period of performance shall be 27 months from the date of execution by the Government. Delivery of services to participants shall commence within 90 days of execution of a grant unless a significant portion of the grant implementation addresses the

development of new curriculum or planning strategies. If enrollments are not anticipated to occur within 90 days, the circumstances should be specifically addressed in the application with the reasons provided and an alternative time frame provided.

8. Option to Extend

DOL may elect to exercise its option to extend these grants for an additional one (1) or two (2) years of operation, based on the availability of demonstration funding under the Workforce Investment Act, successful program operation, and the determination that a grantee's initial program findings could further inform the workforce development system through refinement of the present demonstration.

Part II. Application Process and Guidelines

A. Contents

An original and three (3) copies of the application shall be submitted. The application shall consist of two (2) separate and distinct parts: Part I, the Financial Proposal, and Part II, the Technical Proposal.

1. Financial Application

Part I, the Financial Proposal, shall contain the SF-424, "Application for Federal Assistance". (Appendix A) and the "Budget Information Sheet" (Appendix B). The Federal Domestic Assistance Catalog number is 17.246. The budget shall include on separate pages a detailed breakout of each proposed budget line item found on the **Budget Information Sheet, including** detailed administrative costs and costs for one or more of the following categories as applicable: basic readjustment services, supportive services, and retraining services. The Salaries line item shall be used to document the project staffing plan by providing a detailed listing of each staff position providing more than .05 FTE support to the project, by annual salary, number of months assigned to demonstration responsibilities, and FTE percentage to be charged to the grant. In addition, for the Contractual line item, list each of the planned contracts and the amount of the contract. Where a contract amount exceeds \$75,000, a detailed backup budget to how the amount of the contract was derived must be included. For each budget line item that includes funds or in-kind contributions from a source other than the grant funds, identify the source, the amount, and in-kind contributions,

including any restrictions that may apply to these funds.

Costs associated with the development of curriculum and other one-time costs should be noted separately in order for reviewers to identify costs associated with development and start-up as well as ongoing participant costs.

2. Technical Proposal

Part II, the technical proposal shall demonstrate the offeror's capabilities in accordance with the Statement of Work in Part III of this solicitation. A grant application shall be limited to twenty (20) double-spaced, single-side, 8.5-inch x 11-inch pages with 1-inch margins. Attachments shall not exceed ten (10) pages. Text type shall be 12 point or larger. Applications that do not meet these requirements will not be considered. Each application shall include the Checklist provided as Appendix C, a Time line outlining project activities provided as Appendix D, and an Executive Summary not to exceed two (2) pages. NO COST DATA OR REFERENCE TO PRICE SHALL BE INCLUDED IN THE TECHNICAL PROPOSAL.

B. Hand-Delivered Applications

Applications should be mailed no later than five (5) days prior to the closing date for the receipt of applications. However, if applications are hand-delivered, they must be received at the designated place by 4 p.m., Eastern Time on the closing date for receipt of applications. All overnight mail will be considered to be hand-delivered and must be received at the designated place by the specified time and closing date. Telegraphed and/or faxed proposals will not be honored. Applications that fail to adhere to the above instructions will not be honored.

C. Late Applications

Any application received at the office designated in the solicitation after the exact time specified for receipt will not be considered unless it:

- (1) Was sent by U.S. Postal Service registered or certified mail not later than the fifth calendar day before the closing date specified for receipt of applications (e.g., an offer submitted in response to a solicitation requiring receipt of application by the 30th of January must have been mailed by the 25th); or
- (2) Was sent by U.S. Postal Service Express Mail Next Day Service—Post Office to Addressee, not later than 5 p.m. at the place of mailing two working days prior to the date specified for receipt of application. The term

"working days" excludes weekends and U.S. Federal holidays.

The only acceptable evidence to establish the date of mailing of a late application sent by U.S. Postal Service registered or certified mail is the U.S. postmark on the envelope or wrapper and on the original receipt from the U.S. Postal Service. Both postmarks must show a legible date or the proposal shall be processed as if it had been mailed late. "Postmark" means a printed, stamped, or otherwise placed impression (exclusive of a postage meter machine impression) that is readily identifiable without further action as having been supplied and affixed by an employee of the U.S. Postal Service on the date of mailing. Therefore, applicants should request the postal clerk to place a legible hand cancellation "bull's eye" postmark on both the receipt and the envelope or wrapper.

The only acceptable evidence to establish the date of mailing of a late application sent by "Express Mail Next-Day Service—Post Office to Addressee" is the date entered by the post office receiving clerk on the "Express Mail Next Day Service—Post Office to Addressee" label and the postmarks on both the envelope and wrapper and the original receipt from the U.S. Postal Service. "Postmark" has the same meaning as defined above. Therefore, an applicant should request the postal clerk to place a legible hand cancellation "bull's eye" postmark on both the receipt and the envelope or wrapper.

D. Withdrawal of Applications

Applications may be withdrawn by written notice or telegram (including mailgram) received at any time before award. Applications may be withdrawn in person by the applicant or by an authorized representative thereof, if the representative's identity is made known and the representative signs a receipt for the proposal.

Part III. Statement of Work

Each grant application must follow the format outlined in this Part. For sections A through G below, each application should include:

(1) Information that indicates adherence to the provisions described in Part I, Background (Authorities, Purpose, and Demonstration Policy) and Part II, Application Process and Guidelines, of this announcement; and (2) other information that the applicant believes will address the selection criteria identified in Part IV of this solicitation.

Information required under A and B below shall be provided separately for each labor market area where dislocated workers will be served. To the extent that the project design differs for different geographic areas, information required under section C below shall be provided for each geographic area.

A. Target Population

Describe the characteristics of the proposed target population for the project, e.g., educational level, previous occupation, age range, likely transferable skills, length of unemployment, and language limitations. If that population is limited to one or more subgroups of the dislocated worker population, explain the basis for such limitation. Describe the size and needs of the target population in the local area as they relate to the services available to the grant. Provide documentation showing there is a significant number of dislocated workers with the target population's characteristics in the project area(s). If the project seeks to serve under represented subgroups such as minority groups, women, older workers (50 years of age and older), disabled individuals, within a particular occupation and the selected subgroup has unique characteristics or needs such characteristics or needs should be identified. Substantive and timely documentation of the subgroup's under representation must be included. Note: Up to 5 points of extra credit will be awarded when the targeted population includes at least 40 percent planned enrollment of an under represented subgroup for the occupation in which training will take place.

Indicate how the number of workers to be enrolled was determined. Sufficient documentation should be provided to show that workers with appropriate characteristics to meet the purposes of this grant are available in sufficient numbers to meet the recruitment goals of the grant recognizing that not all workers with appropriate characteristics will chose to participate.

No more than 20 percent of the total demonstration funding allocated by the Department pursuant to this Solicitation for Grant Applications shall be for incumbent workers.

B. Available Jobs

Jobs targeted for this demonstration must be related to the manufacturing industry covered by the Standard Industrial Classification(SIC) Codes 21– 39 and must involve the use of technology skills in a manufacturing setting. Describe the jobs that will be

available and targeted for placement to project participants upon completion of training and placement services including the strategy(ies) for identifying job openings that appear appropriate to the training planned and meet the target wage at placement goals established in the proposal. Include information about the number and type of jobs, wage information and the specific set of skills, knowledge or duties (industry-sponsored standards of certifications). Provide documentation (Footnote sources) that a shortage of qualified workers exists in the local area to fill positions in the targeted occupations in the absence of the proposed project. Anecdotal data should not be used. Information from the Bureau of Labor Statistics (BLS) available through a variety of web sites including BLS, O*NET and America's Labor Market Information System (ALMIS), should be considered as a key source of documentation. In addition, State Occupational Information Coordinating Committee (SOICC) and JTPA Substate Grantee local job training plan may also be considered. Other sources from the private sector such as Chamber of Commerce or local Technology Council surveys as well as university studies are also acceptable. Data must relate to local employment shortages.

Substantive linkages with specific employers who are experiencing skill shortages among their present workforce and/or the demand for additional employees with technology skills in documented occupational shortages must be provided. Letters from employers who have made a commitment to the demonstration project are the most appropriate form of documentation.

If some placements will be made with employers who have not been identified at the time of application, describe the job development and placement strategy to be used to assure placement of demonstration participants.

C. Project Design

(1) Purpose. Describe the specific purpose or purposes of the proposed project.

(Ž) Service Plan. Describe the services to be provided from the time of selection of participants through placement of those participants in jobs. Describe any services to be provided subsequent to job placement. The descriptions shall provide a clear understanding of the services and support that will be necessary for participants to be placed successfully in jobs and to retain those jobs, including services not funded under the grant, and

ways to address participants' financial needs during periods of training. Grantfunded activities should, at a minimum, include recruitment, eligibility determination, assessment, retraining, job placement, and supportive services.

(a) Outreach and recruitment. Describe how eligible dislocated workers will be identified and recruited for participation in the project. Recruitment efforts may address public service communications and announcements, use of media, coordination with the JTPA Service Delivery Area or Substate Grantee, use of community-based organizations and other service groups. Describe the applicant's experience in reaching dislocated workers, especially the targeted population. It is highly recommended that non-JTPA applicants partner with the appropriate JTPA Title III Substate Grantee(s) or local One-Stop Career Center system to plan and implement effective outreach and recruitment strategies.

(b) Eligibility determination. Describe the process to be used in determining the JTPA Title III eligibility of potential participants in the project. It is highly recommended that non-JTPA applicants partner with the appropriate JTPA Title III substate grantee(s) or local One-Stop Career Center system to carry out

eligibility determination.

(c) Selection criteria. Describe the criteria and process to be used in selecting those individuals to be served by the project from among the total number of eligible persons recruited for the project. Explain how the selection criteria relate to the specific purpose of the proposed project. Identify any assessment tools that will be used as

part of selection process.

(d) Training services. Describe the training to be provided—classroom, experiential, on-the-job, internships, etc. Include the length (days and hours) and schedule, any perquisite courses, and customization to account for transferable skills, previous education (note: whether the training requires new and higher educational levels than previous skill training in the same industry), and particular circumstances of the target population and the skill needs of the hiring employer(s). Include information to demonstrate that any proposed training provider is qualified to deliver training that meets appropriate employment standards, and any applicable certification or licensing requirement. Past performance, qualifications of instructors, accreditation of curricula, and similar matters should be addressed if appropriate. Address the costs of proposed training and other services

relative to the costs of similar training and services including courses provided by both public and private providers in the local area. If the training is to be customized to account for individual differences in skills levels of participants or employer hiring needs, describe how these considerations will be taken into account in the delivery of the training.

The training provided must support the information provided regarding skill shortages and demand for jobs using

technology skills.

(e) Job Placement. Describe the role of the employer linkages previously addressed in assuring the availability of jobs for participants completing training. If an Employer Advisory Committee is the primary employer linkage, the members of the committee should be listed and the type of expertise they bring to the committee noted. Provide a discussion of the role(s) of the advisory committee and its projected meeting frequency. Describe any additional job seeking skills training or assistance provided to participants completing training.

(f) Post placement services. Describe any post placement services to be provided and explain their value to the achievement of the project's purpose

and planned outcomes.

(g) Supportive services. Describe those supportive services determined to be appropriate to the target population's needs. Describe policies and procedures to ensure that supportive services are provided only when they are necessary to enable an individual who is eligible for training but cannot afford to pay for such supportive services, to participate in the training program. Indicate how the participants' financial needs during the period of training will be addressed.

(h) Relocation. Describe the limitations and eligibility criteria for relocation assistance, if such assistance

is included in the proposal.

(3) Participant flow. Provide a flowchart noting length of time for various activities (such as one day for assessment, etc.) to illustrate how the project will ensure access to necessary and appropriate reemployment and retraining services. Show the sequence of services and the criteria to be used to determine the appropriateness of specific services for particular participants. Note where service choice options will be available to participants. Indicate the average length of participation from eligibility determination and enrollment in the demonstration project to placement in an unsubsidized job.

(4) Relationship to prior experience. Discuss how the applicant's prior

experience in working with dislocated individuals affects or influences the design of the proposed project. Note especially lessons learned or positive experiences that will be replicated.

D. Planned Outcomes

A description of the project outcomes and of the specific measures, and planned achievement levels, that will be used to determine the success of the project. These outcomes and measures must include, but are not limited to:

(1) The number of participants projected: To be enrolled in services, to successfully complete services through the project, and to be placed into new jobs; a minimum of 80 percent entered employment rate is required;

(2) Measurable effects of the services provided to project participants as indicated by gains in individuals' skills, competencies, or other outcomes;

(3) Wages of participants prior to, at placement and 90 days after placement: (a) For dislocated worker participants: a minimum of 90 percent wage replacement rate is required for at least 75 percent of the participants and an average 90 percent wage replacement for the overall demonstration project is required; (b) for incumbent worker participants: a minimum of 100 percent wage retention is required for all participants successfully completing training and meeting the competencies/ skills levels specified by the employer prior to the training.

(4) For projects serving dislocated workers, as part of the targeted outcome for wage at placement, each project should benchmark at least two key wage averages for the labor market in which each project will operate. Suggested

benchmarks might include:

(a) The average weekly wage in the manufacturing sector, if the project is focused on manufacturing technology; the average weekly wage for technical and skilled trade jobs; or the average weekly wage for computer programmers and (b) the average wage at placement for the JTPA Title III, dislocated worker program operated by the local Substate Grantee. Provide an explanation of the particular benchmarks chosen for the project. For incumbent workers, indicate the present wage level of the workers to be trained and discuss how this wage level compares with the appropriate benchmark wage for the local labor market area.

(5) For each project serving dislocated workers, at least 80 percent of the individuals placed shall be placed at a wage that meets or exceeds (a) the average benchmarked wage in the labor market area, or (b) the average wage at placement for the last program year

completed (currently 1997) for the JTPA Title III dislocated worker program operated by the local Substate Grantee in the targeted labor market, whichever is greater. The manufacturing wage for any labor market may be obtained from the Covered Wages and Employment Program administered by each State's Employment Service.

(6) Customer satisfaction with the project services including participant at critical points in the service delivery process as well as upon placement and employer satisfaction with the skills and preparation of the participants placed

with their organization;

(7) Planned average cost per placement (amount of the grant request divided by the number of program-related placements, and the cost per placement for continued placements (the amount of the grant request minus development/start-up costs divided by the number of program-related placements); and

(8) Other additional measurable, performance-based outcomes that are relevant to the project and which may be readily assessed during the period of performance of the project, such as cost effectiveness of services, comparison with other available service strategies. Where possible, it would also be useful to look at production improvement and other measures the employer uses regarding efficiency, product quality and output.

Note: An explanation of how such additional measures are relevant to the purpose of the demonstration program shall be included in the application.

E. Collaboration

Describe the nature and extent of collaboration and working relationships between the applicant and other workforce development partners in the design and implementation of the proposed project. Include services to be provided through resources other than grant funds under this demonstration. Provide documentation that the collaboration described can reasonably be expected to occur (signed letters of agreement and/or the charter of a formally established advisory council are considered the strongest evidence, while letters of support are considered weaker evidence. Because a core purpose of this demonstration program involves the publicly funded workforce system, the applicant shall describe working relationships with local Substate Grantee(s), including One-Stop/Career Center partners where

Describe the number and types of employers to be directly involved in implementation of the demonstration through activities as participation on an advisory council, provision of input to curriculum development and design, training provider, internship supervision, participation in establishment of local skill standards, etc. Describe activities, presently in place or to be undertaken to link activities to program interventions under this grant to employers, industry, or curriculum/learning centers currently designing and developing occupational/ job skill standards and certifications. Collaboration should focus on linking employers involved in grant activities with any employer, industry, or trade and worker association that has already developed or is developing skill standards certifications. Employer linkages must be specifically addressed in the application and documentation provided of the specific role(s) the employer(s) will play in implementation of the grant provided.

Skill standards play an important role in ensuring participants are meeting the accepted standards of the industry. Grant applicants may show how skills standards and O*NET are used to help dislocated/incumbent workers acquire training and new jobs. Skill standards can mean National Skill Standards (NSS) developed under the auspices of the National Skill Standards Board or other skill standards recognized by employers as valid requirements for jobs. O*NET refers to the Occupational Information Network that replaces the Dictionary of Occupational Titles and defines all jobs in terms of worker requirements, occupational requirements, experience requirements, worker characteristics, occupational characteristics and occupation-specific requirements. The applicant may request a brochure explaining O*NET at the following e-mail address: rannr@doleta.gov.

Skill standards and O*NET are useful for structuring training curriculum, assessing dislocated/incumbent workers' skills and interests, and defining career paths from one occupation to another. Their application in the proposed project's training design would indicate close links to employers and an understanding of the demands faced by workers in high performance workplaces.

Applicants are encouraged to commit matching funds to the implementation and management of their proposed programs. Matches may be in the form of cash or in-kind contributions. These may include but are not limited to such contributions as the development of training modules; payment of tuition costs for training; support for child care

or transportation; and provision of staff time at no cost to the project.

Sources of matching funds may include but are not limited to employers, employer associations, labor organizations, and training institutions. With reference to the sources and amounts of project funds and in-kind contributions identified in the financial proposal as being other than those requested under the grant applied for, describe the basis for valuation of those funds and contributions.

Note: National Reserve Account grants for specific plant closures and layoffs may not be used to match demonstration grant funds, these grants provide sufficient funds to meet the needs of any worker in the targeted dislocation event. However, NRA grant funds may be used to purchase 50 percent or less of the total training slots in a training developed with demonstration grant funds.

Documentation of consultation on the project concept from applicable labor organizations must be submitted when 20 percent or more of the targeted population is represented by one or more labor organizations, or where the training is for jobs where a labor organization represents a substantial number of workers engaged in similar work. Where the union has been involved in bargaining relative to the introduction of either the technology or the addition of new skilled workers at the workplace, provide information as to any role the union played in the design and delivery of the training as well as any impact on the workers with respect to the growth or shrinkage in the number of jobs, the selection of workers for retraining.

F. Innovation

Describe key innovations in the proposed project, including (but not limited to) innovations in concept to be tested, type of participant to be served, services provided, delivery of services, training methods, job development, or job retention strategies. These innovations should be unique to the ongoing knowledge base of service delivery and training presently available to the workforce system. Explain the impact of such innovation on project costs to substantiate the budget items designated as development and start-up costs.

G. Previous Experience

If the applicant has had a demonstration grant with the Department of Labor, Education or HHS within the last three years, list the title of the grant, the amount of the grant, the funding agency, a Federal contact phone number and a brief summary of purpose of the grant. For those grants, funded by

the Department of Labor explain how this grant application differs from grant activity. Explain how the proposed project is similar to and differs from the applicant's prior and current operations.

H. Project Management

(1) Structure. Describe the management structure for the project, including a staffing plan that describes each position and the percentage of its time to be assigned to this project and assures that sufficient staff are available to implement the project in a timely and effective manner. Provide an organizational chart showing the relationship among project management and operational components, including those at multiple sites of the project, in the overall structure of the applicant's organization. Note: It is highly recommended for applicants requesting \$500,000 or more that a full-time project director be available to ensure timely implementation of the project.

(2) Program Integrity. Describe the mechanisms to ensure financial accountability for grant funds and performance accountability relative to job placements, in accordance with standards for financial management and participant data systems in 29 CFR part 95 or 97, as appropriate, and 20 CFR 627.425. Explain the basis for the applicant's administrative authority over the management and operational

components.

Describe how information will be collected to determine the achievement of project outcomes as indicated in section D of this part; and report on participants, outcomes, and expenditures.

(3) Monitoring and Reporting. Describe how the project will keep records of its activities, as required in 29 CFR parts 95 and 97 and 20 CFR 631.63 as appropriate, which will include information such as the following:

(a) Benchmarks. Provide a Time Line of implementation and projected performance benchmarks covering the period of performance of the project (Appendix E). Include a monthly schedule of planned implementation activities and start-up events (such as curriculum development, selection of advisory council, advisory council meetings, hiring of staff, and completion of lease arrangement for space, development of an internal program progress reporting system, design of customer satisfaction measures, initiation of customer satisfaction activities for participants/for employers); quarterly projections of planned participant activity, showing cumulative numbers of enrollments, participation in training and other

services, placements, and terminations; and quarterly cumulative expenditure projections. The quarterly performance projection data may be shown in the same implementation benchmark timeline or separately.

(b) Participant progress. Describe how a participant's continuing participation in the project will be monitored, including determination of successful progress in training activities.

(c) Project performance. Identify the information on project performance that will be collected on a short-term basis (e.g., weekly or monthly) by program managers for internal project management to determine whether the project is accomplishing its objectives as planned and whether project adjustments are necessary.

Describe the process and procedures to be used to obtain feedback from participants, employers, and any other appropriate parties on the responsiveness and effectiveness of the services provided. The description shall identify the types of information to be obtained, the methods and frequency of data collection, and ways in which the information will be used in implementing and managing the project. Grantees may employ focus groups and surveys, in addition to other methods, to collect feedback information. Technical assistance in the design and implementation of customer satisfaction data collection and analysis may be provided by DOL.

(d) Impact of Collaboration and Innovation. Describe the process for assessing and reporting on the impact of collaboration and innovation in the project with respect to the purpose and goals of the demonstration program and the specific purpose and goals of the

project.

(4) Grievance Procedure. If the applicant is a JTPA administrative entity or service provider, assure that a grievance procedure is presently in place. Otherwise, describe the grievance procedure to be used for grievances and complaints from participants, contractors, and other interested parties, consistent with the requirements at Section 144 of JTPA and 20 CFR 631.64(b) and (c).

(5) Previous Project Management Experience. Provide an objective demonstration of the grant applicant's ability to manage the project, ensure the integrity of the grant funds, and deliver the proposed performance. Indicate the grant applicant's past experience in the management of grant-funded projects similar to that being proposed, particularly regarding oversight and operating functions including financial management.

(6) Sustainability and Replicability. Provide assurances that if the project is successful, the demonstration partners will continue to improve and develop the demonstrated approach. Describe the aspects of the demonstration approach that will allow other work force development entities to replicate the proposed project. Note: The cost per participant will be a consideration in any replication consideration by other entities. Discuss the potential applicability of the project, or aspects of the project (such as new assessment tools, etc.), to other dislocated worker programs.

Part IV. Evaluation Criteria

Selection of grantees for awards will be made after careful evaluation of grant applications by a panel selected for that purpose by DOL. Panel results will be advisory in nature and not binding on the ETA Grant Officer. Panelists shall evaluate proposals for acceptability based upon overall responsiveness in accordance with the factors below.

A. Target Population (10 Points)

The description of the characteristics of the target group to be served is clear and meaningful, and sufficiently detailed to determine the potential participants' service need. Documentation is provided showing that a significant number of eligible dislocated workers who possess these characteristics are available for participation within the project area. Sufficient information is provided to explain how the number of dislocated workers to be enrolled in the project was determined. The recruitment plan supports the number of planned enrollments. The target population is appropriate for the specific purpose of the proposed project. The project identifies under represented groups to be trained in the targeted occupation(s).

B. Targeted Jobs (15 Points)

The jobs in the manufacturing industry identified by SIC code are clearly available to workers who have received appropriate training and preparation given:

- (1) The match between the documented skill shortage and the training planned;
- (2) The documentation provided specifying that training meets or is developed based on industry driven skill standards or certifications;
- (3) The substantial level of involvement of employers in making known their needs regarding requisite worker skills necessary for hiring program completers

- (4) The documentation and reliability of job availability is based upon recognized, reliable and timely sources of information
- (5) Where appropriate, the role of workers or representatives of a labor organization representing the workers in the design and/or delivery of training in enhancing worker skills during workplace change

C. Service Plan (12 Points)

The scope of services to be provided is consistent with the demonstration program and project purposes and goals. The scope of services to be provided is adequate to meet the needs of the target population given:

- (1) Their characteristics and circumstances:
- (2) The complexity of the training and the skills to be developed relative to their characteristics and previous job experience
- (3) The jobs in which they are to be placed relative to targeted wage at placement goals;
- (4) The length of program participation planned prior to placement.

D. Costs (20 Points)

Proposed costs are reasonable in relation to the characteristics and circumstances of the target group, the services to be provided, planned outcomes, the management plan, and coordination/collaboration with other entities, including One-Stop/Career Center organizations. The cost information provided regarding similar training available through other training providers is within an acceptable range or sufficient rationale is provided for the cost differences. The impact of development/start-up and innovation on costs is explained clearly in the proposal and is reasonable.

Identification is provided of the specific sources and amounts of other funds which will be used, in addition to funds provided through this grant, to implement the project. The application must include information on any non-JTPA resources committed to this project, including employer funds, grants, and other forms of assistance, public and private. Value and level of external resources being contributed, including employer contributions, to achieve program goals will be taken into consideration in the rating process.

The degree to which other interested partners in the workforce development system invest resources to test the concepts put forth in the application.

D. Management (10 Points)

The project management plan is designed to track project performance in such a way as to assure that benchmarks are achieved in a timely manner, issues affecting performance such as employer involvement, collaboration partners commitments, etc. are quickly identified and addressed, and planned outcomes will be achieved in a cost effective manner.

The applicant (as a part of a collaborative approach) has experience working with technology training. The management structure and management plan for the proposed project will ensure the integrity of the funds requested. The project work plan demonstrates the applicant's ability to effectively track project progress with respect to planned expenditures. Sufficient procedures are in place to use the information obtained by the project operator(s) to take corrective action if indicated. In addition, review by appropriate labor organizations, where applicable, is documented.

The proposal includes a method of assessing customer feedback for both participants and employers involved, and establishes a mechanism to take into account the results of such feedback as part of a continuous system of management and operation of the project.

E. Collaboration (15 Points)

The proposal includes evidence of direct participation by JTPA SubState Grantees and One-Stop/Career Center entities (where present) in the planning and management of this grant. Evidence of participation of employers whose positions are targeted under the grant is present. Evidence of coordination with other programs and entities for project design or provision of services is also be provided. Evidence is presented that ensures cooperation of coordinating entities, as applicable, for the life of the proposed project. The project includes a reasonable method of assessing and reporting on the impact of such coordination, relative to the demonstration purpose and goals and the specific purpose and goals of the proposed project.

F. Innovation (13 Points)

The proposal demonstrates innovation in the concept(s) to be tested, the project's design, and/or the services to be provided. "Innovation" refers to the degree to which such concept(s), design and/or services are not currently found in dislocated worker programs. The project includes a reasonable method of assessing and

reporting on the impact of such innovation, relative to the demonstration program and project purposes and goals.

G. Sustainability and Replicability (5 Points)

The proposal provides evidence that, if successful, activities supported by the demonstration grant will be continued after the expiration date of the grant, using JTPA Title III formula-allotted funds or other public or private resources. The likelihood that the approach may be applicable to a broad range of dislocated worker programs across the country. The proposal provides evidence that the approach and training strategy(ies) used can be replicated by other workforce development partners to address technology skill shortages in their local area.

Grant applications will be evaluated for the reasonableness of proposed costs, considering the proposed target group, targeted jobs, services, outcomes, management plan, and coordination with other entities.

Applicants are advised that discussions may be necessary in order to clarify any inconsistency or ambiguity in their applications. The final decision on awards will be based on what is most advantageous to the Federal Government as determined by the ETA Grant Officer. The Government may elect to award grant(s) without discussion with the applicant(s). The applicant's signature on the Application for Federal Assistance (Standard Form) SF-424 constitutes a binding offer.

Part V. Monitoring, Reporting and Evaluation

A. Monitoring

The Department shall be responsible for ensuring effective implementation of each competitive grant project in accordance with the Act, the Regulations, the provisions of this announcement and the negotiated grant agreement. Applicants should assume that at least one on-site project review will be conducted by Department staff, or their designees. This review will focus on the project's performance in meeting the grant's programmatic goals and participant outcomes, complying with the targeting requirements regarding participants who are served, expenditure of grant funds on allowable activities, collaboration with other organizations as required, and methods for assessment of the responsiveness and effectiveness of the services being provided. Grants may be subject to their

additional reviews at the discretion of the Department.

B. Reporting

DOL will arrange for or provide technical assistance to grantees in establishing appropriate reporting and data collection methods and processes taking into account the applicant's project management plan. An effort will be made to accommodate and provide assistance to grantees to be able to complete all reporting electronically.

Applicants selected as grantees will be required to provide the following reports:

1. Monthly progress reports, during initial start-up and implementation of

the project, and Quarterly Progress Reports.

2. Standard Form 269, Financial Status Report Form, on a quarterly basis.

3. Final Project Report including an assessment of project performance. This report will be submitted in hard copy and on electronic disk utilizing a format and instructions to be provided by the Department.

C. Evaluation

DOL will arrange for or conduct an independent evaluation of the outcomes, impacts, and benefits of the demonstration projects. Grantees must agree to make available records on participants and employers as well as

project financial and management data and to provide access to personnel, as specified by the evaluator(s) under the direction of the Department.

Signed at Washington, DC, this 9th day of December, 1998.

Janice E. Perry,

Grant Officer.

Appendices

- 1. Appendix A—Application for Federal Assistance (Standard Form 424)
- 2. Appendix B— Budget Information Sheet
- 3. Appendix C—Application Checklist
- 4. Appendix D—Implementation Benchmarks and Time Line

BILLING CODE 4501-30-P

Appendix A

APPLICATION FOR

OMB Approval No. 0348-0043

FEDERAL A	ASSISTAN	\mathbf{CE}	. DATE SUBMITTED		Applicant Identifier	
1. TYPE OF SUBMISSION Application □ Construction	Preapplication	3.	DATE RECEIVED BY STA	те	State Application Identifier	
□ Non-Construction	□ Non-Construc	tion 4.	DATE RECEIVED BY FED	ERAL AGENCY	Federal Identifier	
5. APPLICANT INFORMA	;					
Legal Name:				Organizational Unit:		
Address (give city, county, S	State and zip code):			Name and telephone this application (give	number of the person to be contacted or area code):	n matters involving
6. EMPLOYER IDENTIFE	CATION NUMBER (EIN	h:				
				A. State	ICANT: (enter appropriate letter in box H Independent School Dist.	
8. TYPE OF APPLICATIO	□ New □ C	Continuation	□ Revision	B. County C. Municipa D. Township E. Interstate F. Intermunicipal G. Special District	J State Controlled Institution of J. Private University K Indian Tribe L. Individual M. Profit Organization N. Other (Specify):	of Higher Learning
If Revision, enter appropria A. Increase Award D. Decrease Duration	B. Decrease Award Other (specify):	C. Increase	Duration	9. NAME OF FEDE	RAL AGENCY:	
10. CATALOG OF FEDER	AL DOMESTIC ASSIST	TANCE NUMBER:		11. DESCRIPTIVE	TITLE OF APPLICANT'S PROJECT:	
12. AREAS AFFECTED BY	PROJECT (cities, count	ies, States, etc.):				
13. PROPOSED PROJECT	:	14. CONGRESS	IONAL DISTRICTS OF:			
Start Date	Ending Date	a. Applicant			b. Project	
15. ESTIMATED FUNDING	G:		16. IS APPLICATION	SUBJECT TO REVIEW	V BY STATE EXECUTIVE ORDER 12	372 PROCESS?
a. Federal	\$.00			ICATION WAS MADE AVAILABLE T 372 PROCESS FOR REVIEW ON	O THE
b. Applicant	\$.00	DATE	ECUTIVE ORDER 12.	TROCESS FOR REVIEW ON	
c. State	\$.00	b. NO. □ PROGR	AM IS NOT COVEREI	D BY E.O. 12372	
d. Local	\$.00	OR PRO	GRAM HAS NOT BEE	N SELECTED BY STATE FOR REVIE	:w
e. Other	s	.00				
f. Program Income	\$.00	17. IS THE APPLICAN	T DELINQUENT ON	ANY FEDERAL DEBT?	
g. TOTAL	s	.00	□ Yes If "Yes	," attach an explanation	n. 🗆	No
					E TRUE AND CORRECT. THE DOCU IE ATTACHED ASSURANCES IF THE	
a. Typed Name of Authoriz	ed Representative		b. Title			c. Telephone number
d. Signature of Authorized	Representative					e. Date Signed
						1

Previous Editions Not Usable

Standard Form 424 (REV 4-88) Prescribed by OMB Circular A-102

INSTRUCTIONS FOR THE SF 424

This is a standard form used by applicants as a required facesheet for preapplications and applications submitted for Federal assistance. It will be used by Federal agencies to obtain applicant certification that States which are established a review and comment procedure in response to Executive Order 12372 and have selected the program to be included in their process, have been given an opportunity to review the applicant's submission.

Item: Entry:

- 1. Self-explanatory.
- Date application submitted to Federal agency (or State if applicable) & applicant's control number (if applicable).
- 3. State use only (if applicable)
- 4. If this application is to continue or revise an existing award, enter present Federal identifier number. If for a new project, leave blank.
- Legal name of applicant, name of primary organizational unit which will undertake this assistance activity, complete address of the applicant, and name and telephone number of the person to contact on matters related to this application.
- Enter Employer IdentificationNumber (EIN) as assigned by the Internal Revenue Service.
- 7. Enter the appropriate letter in the space provided.
- 8. Check appropriate box and enter appropriate letter(s) in the space(s) provided.
 - "New" means a new assistance award.
 - "Continuation" means an extension for an additional funding/budget period for a project with a projected completion date.
 - "Revision" means any change in the Federal Government's financial obligation or contingent liability from an existing obligation.
- Name of Federal agency from which assistance is being requested with this application.
- 10. Use the Catalog of Federal Domestic Assistance number and title of the program under which assistance is required.
- 11. Enter a brief descriptive title of the project. If more than one program is involved, you should append an explanation on a separate sheet. If appropriate (e.g., construction or real property projects), attach a map showing project location. For preapplications, use a separate sheet to provide a summary description of the project.

Item: Entry:

- 12. List only the largest political entities affected (e.g., State, counties, cities.
- Self-explanatory.
- 14. List the applicant's Congressional District and any District(s) affected by the program or project.
- 15. Amount requested or to be contributed during the first funding/budget period by each contributor. Value of inkind contributions should be included on appropriate lines as applicable. If the action will result in a dollar change to an existing award, indicate only the amount of the change. For decreases, enclose the amounts in parentheses. If both basic and supplemental amounts are included, show breakdown on an attached sheet. For multiple program funding, use totals and show breakdown using same categories as item 15.
- 16. Applicants should contact the State Single Point of Contact (SPOC) for Federal Executive Order 12372 to determine whether the application is subject to the State intergovernmental review process.
- 17. This question applies to the applicant organization, not the person who signs as the authorized representative. Categories of debt include delinquent audit disallowances, loans and taxes.
- 18. To be signed by the authorized representative of the applicant. A copy of the governing body's authorization for you to sign this application as official representative must be on file in the applicant's office. (Certain Federal agencies may require that this authorization be submitted as part of the application.)

Appendix B PART II - BUDGET INFORMATION

SECTION A - Budget Summary by Categories

3. TOTAL Cost Sharing / Match 왕)

(Rate

		(A)	(B)	(C)
1.	Personnel			
2.	Fringe Benefits (Rate %)			
3.	Travel			
4.	Equipment			
5.	Supplies			
6.	Contractual			
7.	Other			
8.	Total, Direct Cost (Lines 1 through 7)			
9.	Indirect Cost (Rate %)			
10.	Training Cost/Stipends			
11.	TOTAL Funds Requested (Lines 8 through 10)			
SECT	ION B - Cost Sharing/ Match	Summary (if a	appropriate) (B)	(C)
1.	Cash Contribution			
2.	In-Kind Contribution			

Use Column A to record funds requested for the initial period of NOTE: performance (i.e. 12 months, 18 months, etc.); Column B to record changes to Column A (i.e. requests for additional funds or line item changes; and Column C to record the totals (A plus B).

INSTRUCTIONS FOR PART II - BUDGET INFORMATION

SECTION A - Budget Summary by Categories

- 1. <u>Personnel:</u> Show salaries to be paid for project personnel.
- 2. Fringe Benefits: Indicate the rate and amount of fringe benefits.
- 3. <u>Travel</u>: Indicate the amount requested for staff travel. Include funds to cover at least one trip to Washington, DC for project director or designee.
- 4. <u>Equipment</u>: Indicate the cost of non-expendable personal property that has a useful life of more than one year with a per unit cost of \$5,000 or more.
- 5. <u>Supplies</u>: Include the cost of consumable supplies and materials to be used during the project period.
- 6. <u>Contractual</u>: Show the amount to be used for (1) procurement contracts (except those which belong on other lines such as supplies and equipment); and (2) sub-contracts/grants.
- 7. <u>Other:</u> Indicate all direct costs not clearly covered by lines 1 through 6 above, including consultants.
- 8. <u>Total, Direct Costs</u>: Add lines 1 through 7.
- 9. <u>Indirect Costs</u>: Indicate the rate and amount of indirect costs. Please include a copy of your negotiated Indirect Cost Agreement.
- 10. <u>Training /Stipend Cost:</u> (If allowable)
- 11. Total Federal funds Requested: Show total of lines 8 through 10.

SECTION B - Cost Sharing/Matching Summary

Indicate the actual rate and amount of cost sharing/matching when there is a cost sharing/matching requirement. Also include percentage of total project cost and indicate source of cost sharing/matching funds, i.e. other Federal source or other Non-Federal source.

NOTE:

PLEASE INCLUDE A DETAILED COST ANALYSIS OF EACH LINE ITEM.

_					٠,			_
Δ:	D.	2	Q	n	α	7	x	С
4 7	\sim	\sim	$\overline{}$		•••	_	2 L	_

Application Checklist

Please complete and submit this checklist with your application. It should be
used as a quick reference of key provisions of the
Solicitation and whether or not these provisions have been included, complied
with or addressed. This document is not intended to be comprehensive or
address every aspect of the solicitation.

Organizacion Applying	•
Contact Person	.•
Phone Number	
Date submitted	
Application Process Please check below: Application is 20 pages or less. _Attachments limited to 10 or fewer. _An original and three copies submitted. _SF424 (Appendix A) included. _Budget Information Sheet (Appendix B) included. _Checklist (Attachment C) included. _Implementation schedule (Attachment D) included. _Executive Summary of two pages or less included.	
Financial and Technical Provisions	
Provide page number below: Target Population identified, with supportive doc	umantation
ranger Population Identified, with Supportive doc	umentation.

Manufacturing industry's(ies') jobs targeted are described and SIC codes are
listed.
Role and involvement in the project of employers experiencing skill
shortages discussed and documented.
Role of the local JTPA Substate Grantee for dislocated worker programs and
One-Stop/Career Center system discussed and documented.
Number and type of targeted jobs and requisite skill sets for employment
addressed.
Specific skill standards and certification for targeted occupations
identified and discussed.
Sources and credibility of labor market/job data cited.
Approach to identifying and recruiting eligible participants included.
Eligibility determination approach discussed.
Process in selecting eligible participants discussed.
Job placement strategy included.
Sequence of services and activities to be provided discussed.
Flowchart of participant services included.
Applicants' prior experience with dislocated workers addressed.
All project outcomes and measures of success specified in Part III D
- A A 3
addressed.
Method of assessing impact of coordination included.
Method of assessing impact of coordination includedCoordination with other entities discussed.
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussed.</pre>
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and</pre>
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships included.
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussed.</pre>
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressed.</pre>
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize data on the project's progress and performance addressed.
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize datā on the project's progress and performance addressedAbility to collect and submit SPIR data indicated.</pre>
<pre>Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize data on the project's progress and performance addressedAbility to collect and submit SPIR data indicatedBenchmarks to indicate planned implementation schedule included.</pre>
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize data on the project's progress and performance addressedAbility to collect and submit SPIR data indicatedBenchmarks to indicate planned implementation schedule includedMethod to obtain feedback from participants and employers
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize data on the project's progress and performance addressedAbility to collect and submit SPIR data indicatedBenchmarks to indicate planned implementation schedule includedMethod to obtain feedback from participants and employers discussed.
Method of assessing impact of coordination includedCoordination with other entities discussedInnovation and impact of the project discussedManagement structure and staffing plan addressedOrganizational chart and relationships includedMechanism to ensure financial accountability discussedBasis for applicant's administrative authority addressedApplicant's Method/System to collect, track, manage, report, and utilize data on the project's progress and performance addressedAbility to collect and submit SPIR data indicatedBenchmarks to indicate planned implementation schedule includedMethod to obtain feedback from participants and employers

Appendix D

IMPLEMENTATION STRATEGY BENCHMARKS AND TIMETABLE FORM MANUFACTURING TECHNOLOGY DEMONSTRATION GRANT DISLOCATED WORKER

can be accomplished effectively and efficiently within the period of performance. The time frames should be realistic ones that project administrators and the all of the necessary tasks to be accomplished and to think through the sequencing of sub-tasks and timing requirements necessary to assure the project may be different and probably includemore tasks than those illustrated below. It is important, however, for any demonstration planning process to set forth Below are examples of the types of tasks found in implementing a demonstration grant. The tasks, methdology and tangible results for any particular project operators can fulfill.

SPECIFIC TASKS TO BE COMPLETED	(for use of grantee not to be included in application submittal) ACCOUNTABLE PERSON	METHODOLOGY/APPROACH TO BE USED	TANGIBLE RESULT	TIME FRAME (begin with Day 1 when notification of grant award is received)	RAME th Day en fon of ard is ed)
	(specify agency/ organization if outside grantee organization)			FROM	5
1. Hire staff		a. Write job description	a. Completed job description b. Announcement published		
		b. Announce opening c. Interview candidates	c. Personnel papers completed on candidate		
		d. Select individual	d. Project at full staff complement		
		e. Individual reports for work			

2. Ensure adequate facilities to operate project	a. Determine space needs b. Arrange for space c. Negotiate lease d. Obtain necessary furniture and other equipment e. Review maintenance and security needs	Lease signed
3. Establish participant reporting procedures	a. Review and identify participant data to be collected b. Form prepared to collect participant data or contract prepared for processing of data c. Staff trained in collection and coding of data d. Management report of data designed and tested	
4. Development of customer satisfaction measures	a. Identify the areas in which to determine customer satisfaction b. With consultant, design and test customer satisfaction survey c. Train staff in customer satisfaction philosophy e. Determine schedule for collection of customer satisfaction data f. Analyze and report on customer satisfaction findings g. Use findings to make appropriate program adjustments	a. Areas identified with regard to participant, employer and training provider satisfaction b & c. Surveys developed and pilot tests conducted d. Staff training completed on customer satisfaction e. Schedule for customer satisfaction data collection established f. Report analyzing customer satisfaction prepared and shared with appropriate staff g. Data used to initiate program adjustments

a. Review required data and report forms to ensure that financial reporting system will collect required information. b. Develop internal report that will ensure that project personnel with operational responsibility will be able to keep informed of project expenditures c. Assign staff responsibility to review reports regularly d	a. Establish responsibility for recruitment results b. Determine appropriate entities to participate in recruitment efforts c. Design recruitment materials	a. Establish contact with b. Develop draft agreements as necessary c. Negotiate and obtain necessary signatures	a. Identify likely employers b. Survey of potential employers to determine need for additional workers and their characteristics c d
5. Develop procedures for collection and reporting of financial data	6. Develop and implement participant recruitment plan.	7. Develop and implement agreements as necessary with other appropriate agencies and entities	8. Develop and implement job development plan

9. Establish training needs, curriculum and select training provider(s)	a. Using information from employers, identify training required for placement with interested employers b. Ensure that curriculum to meet employers' needs is available or develop curriculum in collaboration with employers c. Determine method for selection of training providers	a. Areas most in need of trained workers per area's employers are identified b. Curriculum reviewed with employers c. Training provider(s) selected. d. Any necessary special arrangements with training provider(s) completed.	
10. Establish job search and placement assistance plan for participants	ď		
11. Establish Advisory Panel	a. Identify appropriate members b. Select desired members and issue invitation to participate c. Conduct training of panel regarding responsibilities to project d. Schedule meetings and tentative agendas		

12. Develop	a. Identify data to be used to	ed to
management	determine effectiveness of	
benchmarks for program	project activities (recruitment,	tment,
review and improvement	assessment, training	
	enrollments, training	
	completions, job placements,	nents,
	wage rate, follow-up placement	cement
	rate)	
	b. Determine how data will be	will be
	collected, the frequency of	·of
	review, possible use of data	data
	collected, etc.	